Author's response to reviews

Title: Gender and socioeconomic disparities in BMI trajectories in the Seychelles using birth cohorts generated from serial population-based surveys

Authors:

Isabelle A Rossi (isabelle.rossi@chuv.ch)
Valentin Rousson (valentin.rousson@chuv.ch)
Bharathi Viswanathan (bviswanathan@gov.sc)
Pascal Bovet (pascal.bovet@chuv.ch)

Version: 7 Date: 8 December 2011

Author's response to reviews:

The documents are OK except for the affiliations of the authors.
There are 2 affiliations should appear as follows:

Isabelle Rossi (1), Valentin Rousson (1), Bharathi Viswanathan (2), Pascal Bovet (1,2)

1) Institute of Social and Preventive Medicine (IUMSP), Lausanne University Hospital, Lausanne, Switzerland
2) NCD Section, Department of Public Health, Ministry of Health, Victoria, Republic of Seychelles